Author's response to reviews

Title: Calcified, Minimally Fat-Contained Angiomyolipoma Clinically Indistinguishable from a Renal Cell Carcinoma

Authors:

Jin-Li Chen (j0921713355@yahoo.com.tw)
Sheng-Tang Wu (gu20283@yahoo.com.tw)
En Meng (gu203110@yahoo.com.tw)
Shou-Hung Tang (tansohorn@yahoo.com.tw)
Chih-Wei Tsao (weisurger@gmail.com)
Guang-Huan Sun (ghsun@mail.ndmctsgh.edu.tw)
Dah-Shyong Yu (yuds@ms21.hinet.net)
Sun-Yran Chang (dap77@tpech.gov.tw)
Tai-Lung Cha (tlcha@mail.ndmctsgh.edu.tw)

Version: 3 Date: 19 June 2013

Author's response to reviews: see over
Reply to the reviewer 1:

Thank you for your kind comments. We really appreciate.
Reply to the reviewer 2:

Thank you for the comments and we have made modifications as described below:

1. The pathology picture had been added. Please see Figure 3a and 3b.

2. The plan X-ray showed multiple calcifications in left upper abdomen. IVP was arranged to confirm renal stones or other calcifications, and evaluate the collecting systems and ureteral course. MRI was arranged to differentiate other possibility of renal tumors. We have indicated these tools as auxiliary examinations (at the end of discussion).